Author's response to reviews

Title: Prevalence of arterial stiffness in North China, and associations with risk factors of cardiovascular disease: a community-based study

Authors:

Jinwen Wang (wang-jin-wen@163.com)
Ziqiang Zhou (ziqiang.zhou@gmail.com)
Dayi Hu (dayih0@163.com)

Version: 5 Date: 7 November 2012

Author's response to reviews: see over
Dear Prof Eric Loucks and Miss Marinette Lacson,

According to the outstanding comment of the editor, the authors have stated that age was adjusted for in the paragraph 1 on page 7, paragraph 2 on page 8, and footnotes of Table 6 in this re-revised version of the manuscript.

Many thanks again for the hard work and valuable comments of both editors and reviewers.

Sincerely yours,

Dr. Jin-Wen Wang and

Beijing Anzhen Hospital, Capital Medical University, Beijing Institute of Heart Lung and Blood Vessel Disease, Beijing 100029, People’s Republic of China.
Email: wang-jin-wen@163.com

Dr. Zi-Qiang Zhou

Cardiovascular Center, Beijing Tongren Hospital, Capital Medical University, Beijing 100730, People’s Republic of China.
Email: ziqiang.zhou@gmail.com